Author’s response to reviews

Title: Incidence and associated risk factors for premature death in the Tehran Lipid and Glucose Study Cohort, Iran

Authors:

Ali Eslami (eslami.ali20@gmail.com)
Seyed Sina Naghibi Irvani (sina.irvani@gmail.com)
Azra Ramezankhani (ma.ramezankhani@gmail.com)
Nazanin Fekri (nazanin_fekri@yahoo.com)
Keyvan Asadi (keivanasadi1374@gmail.com)
Fereidoun Azizi (azizi@endocrine.ac.ir)
Farzad Hadaegh (Fzhadaegh@endocrine.ac.ir)

Version: 1 Date: 31 Jul 2018

Author’s response to reviews:

Prof. Natalie Pafitis
Editor in Chief
BMC Public Health

Greetings

Thank you for the opportunity to revise our manuscript (PUBH-D-18-02178). We have carefully revised the entire of manuscript, particularly, abstract, methods and the results sections so that a substantial amount of overlapping information between those sections and our previously published research article (https://doi.org/10.1016/j.ijcard.2016.10.037) has been decreased. Since both studies have been conducted on TLGS, a small amount of overlap between them is unavoidable. We hope that the revised version will meet your expectations.

Thank you for receiving our manuscript and considering it for review. We appreciate your time and look forward to your response.
Yours sincerely

Farzad Hadaegh, MD
Professor of Internal Medicine and Endocrinology
Prevention of Metabolic Disorders Research Center, Research Institute for Endocrine Sciences (RIES), Shahid Beheshti University of Medical Sciences
P.O. Box: 19395-4763, Tehran, Islamic Republic of Iran
Phone: 98 21 22432500
Fax: 22402463
Email: fzhadaegh@endocrine.ac.ir